Author's response to reviews

Title: What do they need to know? Achieving consensus on paediatric musculoskeletal content for medical students.

Authors:

Sharmila Jandial (sharmila.jandial@ncl.ac.uk)
Jane Stewart (jane.stewart@ncl.ac.uk)
Helen E Foster (h.e.foster@ncl.ac.uk)

Version: 4 Date: 11 August 2015

Author's response to reviews: see over
11th May 2015

Dear Mr Ulep,

**MS: 1140850380141868**
*What do they need to know? Achieving consensus on paediatric musculoskeletal content for medical students.*
Sharmila Jandial, Jane Stewart and Helen E Foster

Many thanks for your correspondence dated 7th August 2015 in relation to the above manuscript. I am delighted to enclose a revised manuscript in accordance with your instructions. Please find below a detailed account on how our manuscript has been amended.

1. A full postal address for the corresponding author has been inserted.
2. The abstract has been formatted in line with your guidance and is within the recommended word count (307).
3. Introduction has been renamed as Background.
4. Specific author contributions have been inserted
5. Figure 1, Table 1 and Table 2 have been inserted at the end of the manuscript document.

I hope you will find these changes satisfactory and very much look forward to hearing from you again in the near future. Please let me know if you require any further information.

Yours faithfully,
Dr Sharmila Jandial
Consultant Paediatric Rheumatologist
Great North Children’s Hospital
sharmila.jandial@ncl.ac.uk
Tel 0191 282 5318